Platelet-rich plasma injection for adults with acute Achilles tendon rupture: the PATH-2 RCT

Joseph Alsousou,^{1*†} David J Keene,^{2†} Paul Harrison,³ Philippa Hulley,⁴ Susan Wagland,² Jacqueline Y Thompson,² Scott R Parsons,² Christopher Byrne,⁵ Michael M Schlüssel,⁶ Heather M O'Connor,⁶ Susan J Dutton,⁶ Sarah E Lamb^{2,6} and Keith Willett² on behalf of the PATH-2 study group

¹Institute of Translational Medicine, University of Liverpool, Liverpool, UK ²Kadoorie Centre for Critical Care Research, John Radcliffe Hospital, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

- ³Institute of Inflammation and Ageing, College of Medical and Dental Sciences, University of Birmingham Medical School, University of Birmingham, Birmingham, UK
- ⁴Botnar Research Centre, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK
- ⁵School of Health Professions, Faculty of Health and Human Sciences, University of Plymouth, Plymouth, UK
- ⁶Oxford Clinical Trials Research Unit, Centre for Statistics in Medicine, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

*Corresponding author JosephAlsousou@doctors.org.uk †Joint first authors

Declared competing interests of authors: Joseph Alsousou reports a grant from the National Institute for Health Research (NIHR) Academic Clinical Fellowship programme during the conduct of the study. David J Keene reports grants from a NIHR postdoctoral fellowship during the conduct of the study. Jacqueline Y Thompson reports that she was in receipt of a grant from Thames Valley Clinical Research Network Contingency Fund for physiotherapist training. Sarah E Lamb reports grants from the NIHR Health Technology Assessment (HTA) programme for the AIM study during the conduct of this study. Sarah E Lamb was also a member of the HTA Additional Capacity Funding Board (ongoing), HTA Clinical Trials Board (2010–15), HTA End of Life Care and Add-on Studies (2015), HTA Funding Boards Policy Group (formerly Clinical Study Group) (2010–15), HTA Maternal, Neonatal and Child Health Methods Group (2013–15), HTA post-board funding teleconference (2010–15), HTA Primary Care Themed Call Board (2013–14), HTA Prioritisation Group (ongoing) and NIHR Clinical Trials Unit Standing Advisory Committee (2012–16) during this study. Sarah E Lamb is Co-director of the Oxford Clinical Trials Research Unit and Professor of Rehabilitation at Warwick Clinical Trials Unit, receiving funding from NIHR for both. Sarah E Lamb also receives funding from the NIHR Collaboration for Leadership in Applied Health Research and Care Oxford at Oxford Health NHS Foundation Trust. Keith Willett declares design royalties from Zimmer Biomet (Warsaw, IN, USA), outside the submitted work, for intramedullary bone fixation implants, and held the role of Medical Director for Acute Care, NHS England, during the conduct of this study.

Published November 2019 DOI: 10.3310/eme06120

Plain English summary

The PATH-2 RCT

Efficacy and Mechanism Evaluation 2019; Vol. 6: No. 12 DOI: 10.3310/eme06120

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

A chilles tendon rupture (ATR) is a common injury and leads to months of difficulty with walking. The tendon attaches calf muscle to the heel. Most ATRs in the UK are treated by immobilising the lower leg in a plaster cast or boot, followed by months of exercises to restore calf muscle strength. Absence from work often lasts 2–3 months.

Platelets are the smallest blood cells and contain proteins that promote healing. Platelet-rich plasma (PRP) is a concentrate of a patient's own blood. Laboratory experiments suggest that it could improve tendon healing. The effects of PRP on ATR healing in adults were investigated and recovery using patient-reported measures was measured.

Using a computer, 230 patients from 19 hospitals were randomly allocated to receive either a PRP injection or an imitation injection (placebo). Patients having surgical repair of the tendon were not included. Participants were assessed before treatment and at 4, 7, 13 and 24 weeks after treatment. Information was collected on calf muscle strength, quality of life, pain and whether or not participants recovered the ability to do activities important to them. Any problems with their recovery were monitored. Participants' blood was tested for proteins known to help healing. In 16 participants, tiny samples of tendon tissue were taken to assess the healing.

There were no differences between participants injected with PRP and participants receiving the placebo in calf muscle strength or in the patient-reported measurements. This meant that PRP did not improve tendon healing during the 24 weeks. Complications were similar, with one out of 20 participants in each group having a further tear of the tendon. The number of platelets in PRP did not influence the outcome. The biopsies showed similar healing between the PRP and placebo groups.

It is concluded that PRP does not improve recovery from ATR over 24 weeks. Participants will be reassessed at 2 years. PRP is widely used for other musculoskeletal problems and should be tested just as rigorously in those contexts.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Alsousou *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme support translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 12/206/30. The contractual start date was in October 2014. The final report began editorial review in September 2018 and was accepted for publication in March 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Alsousou *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk